Author's response to reviews

Title: Estrogen independent gene expression defines clinically relevant subgroups of estrogen receptor positive breast cancer

Authors:

Robin M Hallet (halletrm@mcmaster.ca)
John A Hassell (hassell@mcmaster.ca)

Version: 4 Date: 29 October 2014

Author's response to reviews: see over
Dear Editors of BMC Cancer:

We are delighted to hear that our manuscript is suitable for publication in BMC Cancer. We have included responses to your editorial requests below on a point by point basis, which are highlighted in **bold**.

1. 'Please format your abstract according to our guidelines: (http://www.biomedcentral.com/bmccancer/authors/instructions/researcharticle#formatting-abstract).''

   **We reformatted our abstract in accordance with BMC guidelines, it now contains distinct background, methods, results, and conclusion sections.**

2. 'Please move your competing interest section to just before your author contributions section.''

   **We moved the competing interests section so that it precedes the author contributions sections.**

Sincerely yours,

Robin M. Hallett & John A. Hassell